Author's response to reviews

Title: Reporting of harm in randomized controlled trials evaluating stents for percutaneous coronary intervention

Authors:

Morgane Ethgen (morganeethgen@hotmail.com)
Isabelle Boutron (isabelle.boutron@bch.ap-hop-paris.fr)
Philippe Gabriel Steg (gabriel.steg@bch.aphp.fr)
Carine Roy (carine.roy@bch.ap-hop-paris.fr)
Philippe Ravaud (philippe.ravaud@bch.ap-hop-paris.fr)

Version: 7 Date: 16 March 2009

Author's response to reviews:

REPORTING OF HARM IN RANDOMIZED CONTROLLED TRIALS EVALUATING STENTS FOR PERCUTANEOUS CORONARY INTERVENTION

Re: Trials manuscript: 5534201182098351 (T80802P)

We are very grateful for giving us the opportunity to resubmit this manuscript. As we proposed, we updated the analysis to include all trials published after 2005 (i.e., 374 new articles were screened, and 52 new reports were included in the analysis). Despite these modifications, the results were similar.

The manuscript draft remains unchanged, only data were updated. The figure was deleted because will be published in another paper: Quality of Reporting of internal validity and external validity data in Randomized Controlled Trials Evaluating Stents for Percutaneous Coronary Intervention accepted for publication in BMC Medical Research.